

Editorial Office

Baishideng Publishing Group Inc

Type of Article: Original Article - **REBUTTAL LETTER**

Title: A convenient model of hard tissue simulation for dental radiographic research and instruction.

We are sure that the reviewers' comments and suggestions led to the improvement on the manuscript. Therefore, we are very grateful for the contribution provided by the reviewers. We agree with new suggestions and have made the changes accordingly. All modifications are highlighted in yellow as required throughout the manuscript. Below please find our point-by-point responses to the comments and queries of the reviewers.

We look forward to hearing from you,

Reviewer(s)' Comments to Author:

I think the topic of study is interesting. However, I suggest that authors consider the following points:

1. The Introduction section is poorly written.

Thank you for your kindness and detailed considerations. As suggested the Introduction section was improved.

2. I suggest the authors rewrite the Results section.

We thank you very much for the considerations. The Results section was rewrite.

3. References 21-24 are not mentioned in the text.

We are sorry if it's not clear but 21-24 was mentioned at the 4th paragraph of the discussion's topic. Due to new references added, the numbers of this references are now 28-31.

4. The number of keywords is small.

We are sorry for this lapse. The number of keywords was improved.